Author's response to reviews

Title: A new case of de novo 6q24.2-q25.2 deletion on paternal chromosome 6 with growth hormone deficiency: a twelve-year follow-up and literature review

Authors:

Stefano Stagi (stefano.stagi@yahoo.it)
Elisabetta Lapi (elisabetta.lapi@meyer.it)
Marilena Pantaleo (marilena.pantaleo@meyer.it)
Massimo Carella (m.carella@operapadrepio.it)
Antonio Petracca (antonio.petracca@meyer.it)
Agostina de Crescenzo (agostina.decrescenzo@unina2.it)
Leopoldo Zelante (l.zelante@operapadrepio.it)
Andrea Riccio (andrea.riccio@unina2.it)
Maurizio de Martino (maurizio.demartino@unifi.it)

Version: 4 Date: 19 July 2015

Author's response to reviews:

Dear Editor,

we would like to re-submit the article MS: 2140119309133667 “A new case of de novo 6q24.2-q25.2 Deletion on Paternal Chromosome 6 with GH Deficiency: twelve years follow-up and literature review”, revised and modified according to the Editor’ suggestions.

We have created this point-by-point list of corrections according to the Editor’ comments.

Editor's request:

1. Authors have addressed all comments
We thank the Editor for the comment regarding the manuscript

2. Include details of any funding in the Acknowledgement - Please list the source(s) of funding for the study, for each author, and for the manuscript preparation in the acknowledgements section. Authors must describe the role of the funding body, if any, in study design; in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication.

We thank the Editor for the comment. We have modified the Acknowledgment section as for Editor’s suggestions.

3. Hide the identity of the child more fully in Figure 1
We thank the Editor for the suggestions. We have modified the figure 1 as for Editor’s suggestions.